Author’s response to reviews

Title: Choosing the most appropriate existing type 2 diabetes risk assessment tool for use in the Philippines: A case-control study with an urban Filipino population

Authors:

Gina Agarwal (gina.agarwal@gmail.com)
Monserrat Guingona (monsgona@gmail.com)
Jessica Gaber (jgaber@mcmaster.ca)
Ricardo Angeles (angelesric@gmail.com)
Suhasini Rao (raosuh@gmail.com)
Fortunato Cristobal (khryss2448@yahoo.com)

Version: 1 Date: 29 Jan 2019

Author’s response to reviews:

1. Consent to participate
   In the “Ethics Approval and Consent to Participate” section of the Declarations, it is now stated that consent obtained from participants was written.

2. Funding
   In the Funding section, it is now described that the funding body had no role in the design of the study; collection, analysis, and interpretation of data; or writing the manuscript.

3. Abbreviations
   A list of abbreviations is now included before the Declarations section.